Author's response to reviews

Title: Which the specific pathophysiologies is higher in simplified International Index of Erectile Function (IIEF-5) among ED patients?

Authors:

Zhengyan Tang (xytzyan@163.com)
Dongjie Li (jerry1375@126.com)
Xiaobo Zhang (xiaobo5xb@hotmail.com)
Lu Yi (yilu9999@sina.com)
Xiangsheng Zhu (liuying8987@163.com)
Xiangyang Zeng (xytangzy@126.com)
Yuxin Tang (mmcct@126.com)

Version: 3 Date: 26 September 2013

Author's response to reviews:

Sep. 26, 2013
Dear Reymon U. Ampalaya,

Thank you very much for your letter and generous work on our manuscript. Please excuse our delay in replying to your letter. We highly cherish the opportunity given by you to revise our work for possible publication in BMC Urology. We had studied your comments carefully and had made correction which we hope meet with your approval.

First, we made an extensive revision of abstract, and manuscript as well as corrected many typos and grammar errors throughout the text.

Second, we have already stated that “The project organized by the Third Xiangya hospital, Central South University, and it was supported by the other hospitals” in the methods part.

Third, we have revised the Authors' contributions part according to your suggestion carefully

If you still believe that there is something inappropriate in our manuscript, please contact us without hesitation, we would try our best to improve it according to your comment immediately.

We look forward to your response.

On behalf of the authors,

Best regards,

Sincerely,

Yuxin Tang, M.D.,Ph.D.
Department of Urology, the Third Xiangya hospital of Central South University.
Department of Urology, Third Xiangya Hospital of Central South University,